Author’s response to reviews

Title: The Effectiveness of Fermented Turmeric Powder in Subjects with Elevated Alanine Transaminase Levels: a randomised controlled study

Authors:

Sang-Wook Kim (clickm@chonbuk.ac.kr)
Ki-Chan Ha (omphalos@chonbuk.ac.kr)
Eun-Kyung Choi (ekchoi@jbctc.org)
Su-Young Jung (green---tea@hanmail.net)
Min-Gul Kim (mgkim@jbctc.org)
Dae-Young Kwon (dykwon@kfri.re.kr)
Hye-Jeong Yang (yhl@kfri.re.kr)
Min-Jeong Kim (well@kfri.re.kr)
Hee-Joo Kang (anni0216@hanmail.net)
Hyang-Im Back (hyangim100@gmail.com)
Sun-Young Kim (sykim@jbctc.org)
Soo-Hyun Park (shpark@jbctc.org)
Hum-Young Baek (bhy0323@paran.com)
Yong-Jae Kim (yids06@nate.com)
Joon-Yeol Lee (jylee94@nate.com)
Soo-Wan Chae (soowan@jbnu.ac.kr)

Version: 7 Date: 22 January 2013

Author’s response to reviews: see over
Cover letter

Dear Dr Tom Rowles,

MS ID: 1006270834773725

MS title: The Effectiveness of Fermented Turmeric Powder in Subjects with Elevated Alanine Transaminase Levels: a randomised controlled study

Thanks for your kindness and patience.

The authors would like to thank Reviewers (and editors) for careful review of our manuscript and providing us with their comments and suggestions to improve the quality of the manuscript. The following responses have been prepared to address all of the reviewer’s comments.

Reply to the evaluation by referee

1. Are the data sound? (1. Please report the numbers of excluded subjects by group; 2. Please revise the results more systematically)
   
   **Answer 1**: We have described numbers of excluded subjects in both group (p11, line 10, 11).

   **Answer 2**: We have revised result parts (p12, line 6-12).

2. Reviewer’s minor comments

   **Answer**: We have revised our MS in response to reviewer’s minor comments.

We appreciate the comments from the reviewers. Thank you for reviewing our
manuscript.

Sincerely,

Ki-Chan Ha